Author's response to reviews

Title: Expertise About Herbs and Dietary Supplements among Diverse Health Professionals

Authors:

Kathi J Kemper (kkemper@wfubmc.edu)
Paula Gardiner (pgardine@massmed.org)
Jessica Gobble (gobble@wfubmc.edu)
Charles Woods (cwoods@wfubmc.edu)

Version: 3 Date: 3 April 2006

Author's response to reviews: see over
April 3, 2006

Dear Editor:

Re: Expertise About Herbs and Dietary Supplements among Diverse Health Professionals

Major Compulsory Revisions:
None suggested.

Minor Essential Revisions:
As suggested by Dr. Sierpina, we have eliminated the use of the abbreviations, “K, C and C” from the abstract.
We have also changed the first sentence of the Background section as suggested. Thank you.
We changed “Table 5” to “Table 4” on page 7.

Discretionary Revisions
As suggested by Dr. Coeytaux, we have added to the Discussion, “Secular trends do not easily account for the differences in knowledge scores by level of training; it is possible that clinicians acquire some education about HDS as part of practicing their profession.”
We have also expanded the “limitations” paragraph of the Discussion, discussing the potential gender-specific self-selection affecting generalizability of results.